The current issue and full text archive of this journal is available on Emerald Insight at: https://www.emerald.com/insight/1985-2517.htm

# Can effective board drive environmental innovation? The moderating power of CSR committee

Journal of Financial Reporting and Accounting

Muiz Abu Alia and Aladdin Dwekat Accounting Department, An-Najah National University, Nablus, Palestinian Authority

Received 17 May 2024 Revised 8 August 2024 29 October 2024 Accepted 30 October 2024

# Rasmi Meqbel

Department of Accounting, Business School, The Hashemite University, Zarqa, Jordan, and

Tayf Hannoun, Imtiyaz Shakhshir and Aya Naser Accounting Department, An-Najah National University, Nablus, Palestinian Authority

#### Abstract

**Purpose** – This study aims to investigate the impact of various board characteristics on environmental innovation (EI) among companies listed on the STOXX Europe 600. It also examines the moderating role of CSR committees on the board–EI nexus.

**Design/methodology/approach** – The sample consists of companies listed on the STOXX Europe 600 index over 12 years (2011–2022). This study uses the Refinitiv Eikon database to evaluate the extent of EI. Panel data regression analysis is used, with two-stage least squares and lagged models used as robustness tests to control for endogeneity.

**Findings** – The results indicate that board independence and gender diversity significantly increase EI, whereas CEO duality negatively impacts it. Other board attributes, such as board size, show no impact on EI. In addition, the presence of CSR committees moderates these relationships, enhancing the positive effects of gender diversity and board independence and mitigating the negative impact of CEO duality.

**Practical implications** – This study provides valuable insights for policymakers and corporate strategists aiming to advance environmental responsiveness through strategic board composition and establishing CSR committees. Emphasizing the importance of board independence, gender diversity and CSR committees, the findings suggest practical pathways for enhancing the adoption of EI by creating governance structures that support sustainable practices.

**Originality/value** — To the best of the authors' knowledge, this is the first study to examine the moderating role of CSR committees on the associations between board characteristics and EI. This research addresses a crucial gap in the current literature, enriching the understanding of corporate governance and sustainability. It provides critical insights for developing policies and strategies that promote EI through effective board composition and the implementation of CSR committees.

**Keywords** Environmental innovation, Corporate governance, Board characteristics, CSR committees, STOXX Europe 600

Paper type Research paper



#### 1. Introduction

Environmental innovation (EI) is increasingly crucial for sustainable development due to rising environmental degradation and heightened stakeholder awareness of corporate ecological responsibilities (Mohamed Riyath and Inun Jariya, 2024; Ahmed *et al.*, 2024;

Journal of Financial Reporting and Accounting © Emerald Publishing Limited 1985-2517 DOI 10.1108/JFRA-05-2024-0280

Brunnermeier and Cohen, 2003). As business operations grow and resources become limited, the pressure to balance economic objectives with environmental preservation intensifies, motivating companies to pursue innovations that promote sustainability while meeting stakeholder demands (Estapé-Dubreuil *et al.*, 2016; Hunjra and Hussainey, 2024; Abweny *et al.*, 2024; Alia *et al.*, 2024). These innovations, encompassing advancements in processes, products, corporate practices and marketing, integrate sustainability into business operations, addressing ecological issues while enhancing economic growth, corporate reputation and competitive edge (Nandy *et al.*, 2024). Despite these benefits, EI adoption faces obstacles such as financial risks, lack of data, funding shortages and uncertainty, which often reinforce traditional business models. These challenges emphasize the need for supportive regulations, strategic management and organizational change to enable the shift toward innovative and sustainable business practices (Yin and Wang, 2018; Ruiz-Castillo *et al.*, 2023).

At the microlevel, the success of EI largely centers on how effectively management integrates sustainable practices into business strategies (Farza *et al.*, 2022). Corporate Governance (CG), mainly through the board of directors, is instrumental in guiding organizational behavior toward environmental sustainability (Abdelhaq *et al.*, 2024). The board's strategic influence extends to setting social and environmental goals, enhancing sustainability efforts, promoting innovation and managing EI-related challenges (Nadeem *et al.*, 2020). Research indicates that gender diversity on boards significantly enhances EI (Galia *et al.*, 2015; Nadeem *et al.*, 2020; Farza *et al.*, 2022), and independent directors provide valuable expertise crucial for these initiatives (García-Sánchez *et al.*, 2021). In addition, firms with larger boards are more likely to adopt environmentally sustainable practices (Almaqtari *et al.*, 2023).

CSR committees (CSRCs) have emerged as crucial governance mechanisms that strengthen a firm's environmental performance, demonstrating a commitment to sustainability (Gennari and Salvioni, 2019; Velte and Stawinoga, 2020). These committees assist boards in addressing social and sustainability challenges more effectively, enhancing engagement and accountability in managing CSR strategies (Baraibar-Diez and Odriozola, 2019; Pucheta-Martínez and Gallego-Álvarez, 2019). CSRCs thus play a crucial role in reinforcing and executing the board's environmental strategies, strengthening the firm's overall approach to sustainability.

Research extensively explores environmental governance, particularly examining how corporate board characteristics influence environmental outcomes like environmental disclosure (Khaireddine *et al.*, 2020), performance (García Martín and Herrero, 2020; Khan *et al.*, 2021) and sustainable production practices (Almaqtari *et al.*, 2023). However, there is limited research on the impact of board characteristics, specifically on EI (Farza *et al.*, 2022; Konadu *et al.*, 2022; García-Sánchez *et al.*, 2021; Nadeem *et al.*, 2020; Galia *et al.*, 2015). These studies often focus on specific board attributes within narrow geographical or sectoral contexts. For instance, Farza *et al.* (2022) analyzed board independence and gender among German HDAX firms, whereas Konadu *et al.* (2022) and Nadeem *et al.* (2020) studied gender diversity in US-listed firms. García-Sánchez *et al.* (2021) examined global agri-food firms, and Galia *et al.* (2015) focused on French-listed companies, assessing gender and age diversity.

There are inconsistencies in findings, particularly concerning the influence of female directors on EI. While some studies (Wang *et al.*, 2022; Farza *et al.*, 2022; Nadeem *et al.*, 2020) report a positive effect, others (Agustia, 2023; Traversi *et al.*, 2024) present conflicting results. In addition, a significant gap exists in the literature, as no studies have explored the moderating effects of CSRC on the relationship between board characteristics and EI.

Accordingly, this study aims to bridge this significant gap by evaluating the impact of various board characteristics – including board size, independence, gender diversity and CEO duality (CEOD) – on the EI of firms listed in the STOXX Europe 600 from 2011 to 2022. Moreover, this research examines whether the relationship between board characteristics and EI is influenced by the presence of CSRC.

This study significantly contributes to the literature on the CG–EI nexus, a topic previously explored with mixed results, often focusing on specific attributes within certain national contexts (e.g. Farza *et al.*, 2022; Nadeem *et al.*, 2020; Galia *et al.*, 2015). By analyzing firms listed on the STOXX Europe 600, this research extends its geographic reach and enhances its depth by incorporating the European regulatory context, including frameworks like the NFRD (Velte, 2024). Covering data from 2011 to 2022, the study offers a broader perspective across 17 European countries, contrasting with prior single-country analyses. In addition, it introduces novel insights by examining the moderating role of CSRC in the relationship between board characteristics and EI, highlighting their importance in executing effective environmental strategies. These findings provide practical implications for CSR specialists, corporate policymakers and governance bodies, offering guidance on optimizing board structures and CSRC functionalities to enhance EI and align with current European regulatory standards.

The paper is structured as follows: Section 2 outlines the hypotheses within the context of the theoretical framework and relevant literature. Section 3 details the methodology, including data collection, sample selection, variable measurement and regression models used. Section 4 presents the empirical results, covering descriptive statistics, correlation analysis and regression outcomes. Finally, Section 5 discusses the implications of the findings and offers suggestions for future research.

## 2. Theoretical framework and hypotheses development

#### 2.1 Theoretical framework

The rising consumer awareness of the environmental impact of corporate activities has compelled businesses to prioritize sustainability initiatives (Abdelhaq and Dwekat, 2024; Alkaraan, 2022; Dwekat *et al.*, 2020b). Companies are now incorporating long-term strategies that integrate innovative processes, products and organizational structures, aligning their operations with stakeholder values (Rahi *et al.*, 2023). These efforts aim not only to minimize environmental damage but also to enhance profitability and competitiveness (Nadeem *et al.*, 2020). Despite these efforts, implementing EI remains complex and costly (Del Río *et al.*, 2016).

EI requires committed leadership and strong organizational structures, with boards playing a critical role in setting strategic objectives and ensuring the implementation of effective environmental management systems (Hussainey *et al.*, 2022; Wagner, 2008). This study uses Agency, Stakeholder, Legitimacy and Resource Dependency Theories to investigate how board characteristics influence EI.

Agency Theory highlights the necessity for governance mechanisms that align managerial decisions with shareholder interests, especially given the risks associated with innovation (Xue *et al.*, 2022; Kayed and Meqbel, 2024; Alta'any *et al.*, 2024). Boards with diverse, skilled members, particularly those with a CSRC, are better positioned to manage these risks and strengthen oversight (García-Sánchez *et al.*, 2019a; Moreno-Ureba *et al.*, 2022). CSRCs are essential in balancing short-term profitability with long-term value creation and guiding environmental policies (Gennari and Salvioni, 2019; Dixon-Fowler *et al.*, 2017).

Stakeholder Theory promotes integrating various stakeholder interests into CG (Voinea *et al.*, 2022). Boards with sufficient expertise can effectively manage sustainability and innovation to meet stakeholder expectations, enhancing corporate reputation (Gennari, 2019; Ginesti *et al.*, 2023). The CSRC's multidisciplinary nature supports these efforts by formulating and evaluating CSR strategies (Velte and Stawinoga, 2020; Orazalin, 2020).

Legitimacy Theory suggests that meeting societal expectations is essential for firms to sustain their operations. As societal values evolve toward environmental consciousness, establishing CSRCs can help firms gain legitimacy and enhance their reputation (Gennari and Salvioni, 2019; Velte and Stawinoga, 2020).

Resource Dependency Theory highlights that board diversity is crucial for accessing resources necessary for strategic development (Uyar *et al.*, 2020). Boards with diverse expertise and networks can secure essential resources, enabling the company to exceed environmental standards and implement effective initiatives (Hillman *et al.*, 2009). CSRCs provide specialized knowledge to support environmental efforts (Dixon-Fowler *et al.*, 2017).

## 2.2 Hypotheses development

The board of directors oversees management activities, including those related to social responsibility and sustainability (Ramdhony *et al.*, 2023; Alkaraan *et al.*, 2024). The board's composition, structure and activity level indicate a firm's commitment to CSR and its approach to environmental and social concerns (Uyar *et al.*, 2020). High board efficiency and qualifications enhance its capacity to fulfill its responsibilities (García-Sánchez *et al.*, 2020; Rossi *et al.*, 2021; Konadu *et al.*, 2022; Wu *et al.*, 2023). Wang *et al.* (2022) suggested that an efficient board supervises management effectively, curbing opportunistic behavior, improving decision-making and encouraging innovation. Moreover, an efficient board provides essential information and resources, reducing uncertainties associated with innovation (Lu and Wang, 2018).

2.2.1 Board gender diversity. Board diversity, especially the inclusion of female directors, is widely recognized as a critical driver of improved firm performance. Diverse boards reduce agency costs and enhance oversight, contributing positively to firm outcomes (Alqatan, 2024; Konadu *et al.*, 2022). Female directors bring creativity, skills and a solid commitment to legal and ethical standards, strengthening stakeholder relationships and improving a company's reputation and competitive position (Rahman *et al.*, 2024). Their involvement is particularly impactful in environmental performance, as they contribute effectively to decision-making and developing environmental strategies (Elad Fotoh *et al.*, 2018; Nadeem *et al.*, 2020).

Empirical evidence supports that gender-diverse boards are more effective at implementing EI (Alia and Mardawi, 2021). Studies link higher female representation with enhanced EI (Wang *et al.*, 2022; Farza *et al.*, 2022). However, some research (Agustia, 2023; Traversi *et al.*, 2024) indicates that gender diversity may negatively influence EI under certain conditions, highlighting the complexity of managing board dynamics.

According to the Resource Dependency Theory, diverse backgrounds and expertise among board members, especially female directors, reduce a firm's resource dependence and improve performance. Female directors' ethical and cautious nature supports effective board monitoring, aligning decisions with shareholder interests (Issa and Bensalem, 2023). Diversity is crucial for managing complex issues like EI, as it helps boards integrate societal and environmental interests into corporate strategies (Miller and del Carmen Triana, 2009). Female directors' innovative and collaborative problem-solving skills enhance the board's capacity to develop and implement effective EI strategies:

*H1*. Board Gender Diversity positively affects the environmental innovation of the companies included in the STOXX Europe 600 Index.

2.2.2 Board size. According to Resource Dependency Theory, larger boards offer more resources, expertise and connections that help firms reduce external dependencies and boost their innovative capabilities (Chouaibi and Jarboui, 2012). This diversity is crucial for managing the complexities and risks associated with environmental sustainability initiatives.

Stakeholder Theory aligns with this perspective, suggesting that larger boards are better positioned to accommodate diverse stakeholder interests, which is essential for companies tackling environmental challenges (Freeman *et al.*, 2010; Freeman and Evan, 1990). Members with direct links to key environmental stakeholders – such as regulatory bodies, NGOs and community groups – can help integrate these perspectives into the firm's innovation strategies (Lacetera, 2001; Shapiro *et al.*, 2015). This inclusion ensures that environmental initiatives are comprehensive and aligned with societal goals.

Empirical studies demonstrate the advantages of larger boards due to their diverse expertise, which is critical for decision-making in EI (Cheng, 2008; Fuente *et al.*, 2017). Research shows that larger boards enhance strategic programming and governance effectiveness, leading to improved environmental outcomes (Fuente *et al.*, 2017; Moreno-Gómez *et al.*, 2017). In addition, diverse boards are more likely to develop creative solutions to environmental challenges (Miller and del Carmen Triana, 2009).

However, Agency Theory highlights potential drawbacks, such as increased coordination costs and conflict (Jensen, 1993; Goodstein *et al.*, 1994). These challenges can be mitigated through structured governance practices that enhance coordination, ensuring the benefits of larger boards without inefficiencies. The hypothesis is thus formulated based on these insights:

- *H2*. Board size positively affects the environmental innovation of the companies included in the STOXX Europe 600 Index.
- 2.2.3 Board independence. Board independence is vital to CG, particularly in the Anglo-American context where ownership is widely dispersed. Agency theory suggests that independent directors play a crucial role in reducing conflicts between shareholders and management by curbing managerial opportunism and ensuring executives prioritize stakeholders' interests (Lu and Wang, 2018; Zhu et al., 2022). Beyond shareholders, Stakeholder theory highlights that independent directors protect a wider group of stakeholders, ensuring adherence to ethical practices and social responsibilities essential for long-term success. Empirical studies, such as those by Ntim and Soobaroyen (2013) and García-Sánchez et al. (2020), demonstrate that independent directors enhance management oversight, promoting social responsibility and sustainable development.

Resource Dependency Theory further argues that independent directors provide essential resources, including expertise, information and external networks, vital for innovation and strategic renewal. Studies show that having board members focusing on environmental issues significantly improves a firm's environmental practices (Khaireddine *et al.*, 2020; Osemene *et al.*, 2021). The positive influence of independent directors on EI is supported by research indicating that these directors improve a firm's vision and responsiveness to environmental needs, promoting sustainable innovation (Farza *et al.*, 2022; Lu and Wang, 2018). Based on this evidence, the hypothesis was formulated:

*H*3. Board independence positively affects the environmental innovation of the companies included in the STOXX Europe 600 Index.

2.2.4 CEO duality. CEOD presents significant governance challenges, particularly regarding EI, as it often results in a concentration of power that weakens board independence and reduces oversight effectiveness in sustainable practices (Krause *et al.*, 2013; Ghardallou, 2022). Agency Theory advocates for separating the roles of CEO and chairman to maintain the necessary independence for effective governance (Korir and Tenai, 2020). This separation ensures that decisions are impartial and align with the long-term interests of shareholders and stakeholders, including environmental priorities (Rossi *et al.*, 2021).

Research suggests that CEOD tends to prioritize financial performance, potentially at the expense of broader social and environmental objectives, which are essential for sustainable corporate strategy (Zhang, 2012). Studies indicate that CEOD can hinder sustainable performance by limiting governance's role in advancing EI (Zhu *et al.*, 2022). It can also stifle contributions from diverse board members, such as female directors, who typically positively influence ESG outcomes (Romano *et al.*, 2020; Ghardallou, 2022). Evidence from Mallin and Michelon (2011) shows that sustainability performance declines under CEOD due to conflicts of interest and a lack of long-term environmental focus.

While CEOD may streamline decision-making, empirical studies highlight that this efficiency often sacrifices the comprehensive and deliberative approach needed for EI (Meckling and Jensen, 1976). The dual leadership structure may lead to a focus on short-term financial gains rather than long-term investments in EI, which are critical for sustainable development (Rossi and Cebula, 2015). Based on these insights, the hypothesis is formulated:

*H4*. CEO duality negatively affects the environmental innovation of the companies included in the STOXX Europe 600 Index.

# 2.3 The moderating effect of CSR committee

The formation of a CSRC demonstrates a company's proactive approach to sustainability and CSR, addressing growing stakeholder expectations and the complexities associated with CSR challenges (Spitzeck, 2009; Eberhardt-Toth, 2017). This strategic initiative enhances the firm's legitimacy and effectively shows its commitment to managing social and environmental responsibilities (Godos-Díez *et al.*, 2018; Mallin and Michelon, 2011). CSRCs are vital in advancing environmental and social initiatives, positively influencing eco-innovation and improving governance effectiveness. By overseeing environmental decisions, they manage uncertainties related to strategic actions, ensuring better outcomes (Moreno-Ureba *et al.*, 2022). In addition, CSRCs enhance the management of environmental issues through specialized oversight (Dixon-Fowler *et al.*, 2017; Radu and Smaili, 2021) and guide CSR strategies to create competitive advantages while raising environmental awareness, ultimately reducing negative impacts like emissions (Konadu *et al.*, 2022; Gennari and Salvioni, 2019). This committee supports directors in developing and monitoring CSR strategies, thereby strengthening the board's capacity in these areas (Velte and Stawinoga, 2020; Javeed *et al.*, 2022; Dwekat *et al.*, 2022a).

Empirical studies emphasize the influence of CSRCs on corporate environmental performance. For instance, Pucheta-Martínez and Gallego-Álvarez (2019) found that CSRCs significantly enhance CSR policy transparency and effectiveness, leading to improved stakeholder trust and corporate reputation. Similarly, Endrikat *et al.* (2021) demonstrated that CSRCs align board characteristics with CSR objectives, enhancing overall sustainability practices. Furthermore, CSRCs facilitate decentralized decision-making within the board, enabling targeted management of CSR issues, which improves strategic development, boosts corporate accountability and strengthens the firm's legitimacy and reputation by managing

stakeholder relations and reducing information asymmetries and agency conflicts (García-Sánchez *et al.*, 2019a; Dwekat *et al.*, 2022b).

Qaderi *et al.* (2022) provided evidence that CSRCs amplify the effectiveness of board characteristics on CSR reporting, acting as essential governance mechanisms that increase the demand for CSR assurance. Baraibar-Diez and Odriozola (2019) noted that CSRCs positively influence the link between CSR-related compensation and CSR outcomes, indicating their strategic role in aligning executive compensation with CSR results for improved corporate performance.

Thus, the CSRC is an integral component of modern CG, ensuring that environmental and social policies are effectively implemented, leading to substantial benefits. Based on these arguments regarding the CSRC's impact on board efficiency and CSR performance, the final hypothesis is formulated:

*H*5. The presence of CSRC moderates the relationships between board characteristics and environmental innovation.

## 3. Methodology

# 3.1 Sample selection and data sources

This study draws on a data set comprising firms from the STOXX Europe 600 Index, covering the period 2011–2022, resulting in 6,126 firm-year observations. The STOXX Europe 600 Index includes companies of varying market values (large, medium and small) from 17 European countries, representing approximately 90% of the region's freely traded stock market value. The UK dominates the index, accounting for around 28%, followed by France, Germany and Switzerland, each contributing roughly 15%.

Several factors support the choice of the STOXX Europe 600 Index. European firms are recognized for their commitment to environmental sustainability, influenced by the region's advanced environmental policies and the emphasis on sustainable practices across industries (European Commission, 2021). The European environmental research and innovation policy encourages collaborative efforts across sectors to promote sustainable and economically viable solutions (Mongo *et al.*, 2021). In addition, the increasing demand for eco-friendly products has prompted companies to adopt innovative strategies that enhance product quality while meeting environmental regulations (Song *et al.*, 2020).

The selection of this index is further justified by its broad geographic and industrial representation, encompassing firms from 17 countries with diverse regulatory frameworks, environmental policies and market dynamics. This diversity creates a comprehensive environment to examine the impact of board characteristics on EI. Excluding financial companies due to their unique regulatory conditions (Albitar and Hussainey, 2023; Dwekat *et al.*, 2022a) refines the analysis, allowing for a more focused study of the European corporate sector's board characteristics and EI relationship. Additionally, the robust regulatory landscape shaped by directives such as the NFRD, CSRD, EU Taxonomy Regulation and the proposed CSDD Directive enhances the relevance of this index for this research (Velte, 2024). Previous studies have also used the STOXX Europe 600 Index in CG and CSR research, supporting its appropriateness (Dwekat *et al.*, 2022a; Velte, 2024; Belhouchet and Chouaibi, 2024).

# 3.2 Variables measurement

*3.2.1 Dependent variable.* This study measures EI as the dependent variable using the EI Score (EIS) from the Eikon database. The EIS, which ranges from 0 to 100, assesses a

company's ability to develop and improve environmental technologies and processes. A higher score, closer to 100, indicates a greater capacity for EI, demonstrating the firm's effectiveness in minimizing environmental costs and impacts. This metric provides reliable, auditable data, reduces reliance on self-reported information and minimizes replication errors, improving generalizability. In addition, the EIS highlights a firm's ability to gain competitive advantages through the introduction of eco-designed product results or processes, as noted in previous studies (Nadeem *et al.*, 2020; Zaman *et al.*, 2021; Moreno-Ureba *et al.*, 2022). This measurement evaluates how firms manage financial and environmental challenges while enhancing market opportunities.

3.2.2 Independent, moderating and control variables. This study includes key board characteristics such as gender diversity (BOGE), board size (BOSI), board independence (BOIN) and CEOD, which are consistent with previous research. BOGE is measured as the proportion of female directors on the board (Nadeem *et al.*, 2020; Farza *et al.*, 2022). BOSI represents the total number of directors (Godos-Díez *et al.*, 2018; Wang *et al.*, 2019). BOIN is the percentage of independent directors, potentially enhancing EI through stronger oversight (García-Sánchez *et al.*, 2019b; Omran *et al.*, 2021). CEOD is captured as a binary variable where a value of 1 indicates the CEO also serves as chairman, centralizing decision-making (Krause *et al.*, 2013; Nadeem *et al.*, 2020).

Control variables include firm size (SIZE), measured by the natural logarithm of total assets, as larger firms typically have more resources for EI (Nadeem *et al.*, 2019; Farza *et al.*, 2022). Financial leverage (LEV) is the ratio of total debt to total assets, reflecting the firm's financial strategy, which may influence its EI investment capacity (Carmona *et al.*, 2022; Dwekat *et al.*, 2022a). Return on equity (ROE) is also considered, assuming that more profitable firms are better positioned to invest in EI (Nadeem *et al.*, 2019; Kayed *et al.*, 2024).

A CSRC is included as a binary variable, as it indicates a structured approach to CSR, likely promoting EI by prioritizing environmental issues (Albitar *et al.*, 2024; Mardawi *et al.*, 2023). Table 1 summarizes all study variables, providing their definitions and measurement methods.

#### 3.3 Regression model

The following multiple regression model is used to estimate the impact of the characteristics of board of directors on the EI:

$$EIS = \alpha_0 + \beta_{1BOGE} + \beta_{2BOSI} + \beta_{3BOIN} + \beta_{4CEOD} + \beta_{5SIZE} + \beta_{6ROE} + \beta_{7LEV} + \beta_{8CSRC} + \varepsilon_{it}$$

Table 1 identifies the dependent, independent and control variables;  $\beta$ k are the regression coefficients and  $\varepsilon$  is the error term or regression residual.

#### 4. Results

## 4.1 Descriptive and correlation

Table 2 summarizes the descriptive statistics for the study variables. The average EIS is approximately 41%, indicating varying levels of corporate engagement in environmental initiatives. The average board size is 11 members, with 61% being independent directors and 26% women. Around 21% of the firms exhibit CEOD, where the CEO also serves as the board chair.

Regarding control variables, the average firm size, measured as the natural logarithm of total assets, is roughly 15.599. The mean financial leverage is 0.219, reflecting the typical debt-to-asset ratio. A CSRC is present in about 64% of firms, suggesting a widespread

Table 1. Study variables

Variable	Label	Measurement
Dependent variable		
Environmental innovation	EIS	Environmental Innovation Score (EIS), ranges from 0 to 100. The closer the score to 100, the higher the environmental innovation is
Independent variables		
Board gender diversity	BOGE	Female directors' percentage (number of female directors to the total number of the board members)
Board size	BOSI	The number of board members
Board independence	BOIN	The percentage of independent directors on the board
CEO duality	CEOD	A dummy variable (1 if the CEO is also the chairman of the board, 0 otherwise)
Control variables		
Firm size	SIZE	The natural algorithm of the total assets
Return on equity	ROE	Net income to total equity
Financial leverage	LEVE	Total debt to total assets
CSR committee	CSRC	A dummy variable (1 if CSR committee exists, 0 otherwise)
Source: Authors' own work		

Table 2. Descriptive statistics

Variable	Mean	Std. dev.	Min	Max
EIS	40.634	31.698	0	99.82
BOGD	26.998	12.78	0	65.64
BOSI	10.965	3.762	4	21
CEOD	0.212	0.494	0	1
BOIN	61.259	22.452	0	100
SIZE	15.599	1.335	12.893	17.818
LEVE	0.215	0.185	0	0.501
ROE	14.195	14.108	-16.05	46.52
CSRC	0.636	0.481	0	1
Source: Authors	Source: Authors' own work			

emphasis on structured CSR governance. The average ROE stands at 14.195%, indicating overall financial performance among the sampled firms.

Pearson's correlation coefficients, presented in Table 3, evaluate multicollinearity among the study variables. The highest observed correlation is 0.51 between BOSI and SIZE. Correlation coefficients between other variables show low to moderate relationships, all below the critical threshold of 0.70. These results indicate that multicollinearity is not a concern for testing the study's hypotheses.

# 4.2 Regression analysis

Table 4 presents the regression results from two models: Model 1 (pooled OLS) and Model 2 (Fixed Effects). Continuous independent variables are winsorized at the 1st and 99th

**Table 3.** Correlation matrix

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
(1) EIS (2) BOGD (3) BOSI (4) BOIN (5) CEOD (6) SIZE (7) LEV (8) ROE (9) CSRC	1,000 0.148* 0.221* 0.081* -0.072* 0.363* -0.063* -0.031 0.2429*	1.000 0.120* 0.174* -0.074* 0.165* -0.004 0.029 0.1191*	1.000 -0.304* -0.213* 0.512* 0.017 -0.015 0.2419*	1.000 0.123* 0.050* 0.042* 0.005 0.1010*	1.000 -0.130* 0.037 0.020 -0.0896*	1.000 0.112* -0.103* 0.4115*	1.000 0.023 -0.0302	1.000 -0.0303	1

**Notes:** \*\*\*p < 0.01; \*\*p < 0.05; \*p < 0.1

Source: Authors' own work

**Table 4.** The impact of board characteristics on firms' environmental innovation

VARIABLES	(1) Innovation	(2) Innovation
BOGD	0.173***(0.0389)	0.166***(0.0414)
BOSI	5.366*(1.776)	6.418(1.695)
BOIN	0.0898***(0.0229)	0.104***(0.0217)
CEOD	-1.879**(1.109)	-3.277***(1.087)
SIZE	7.143***(0.451)	6.616***(0.441)
LEV	-27.69***(3.552)	-40.38***(3.623)
ROE	0.00689**(0.00563)	0.00918*(0.00583)
CSRC	9.686***(1.194)	9.570***(1.161)
Constant	-94.40***(6.094)	-92.67***(6.388)
Year fixed effects	No	Yes
Country fixed effects	No	Yes
Industry fixed effects	No	Yes
Observations	6,126	6,126
Adjusted R-squared	0.161	0.229

**Notes:** The table presents regression model results for European companies listed on the STOXX 600 index from 2011 to 2022. Model 1 uses pooled OLS to analyze the impact of board characteristics on EI, whereas Model 2 incorporates controls for industry, year and country to address time-invariant industry differences and macroeconomic variations. Robust standard errors, adjusted for heteroscedasticity and clustered at the firm and industry levels, are applied to improve accuracy. Variable definitions are provided in Table 1. Statistical significance levels are indicated as follows: \*\*\* for 1%, \*\* for 5% and \* for 10%

Source: Authors' own work

percentiles to control for extreme values (Dhaliwal *et al.*, 2011). To address heteroscedasticity, robust standard errors are applied (Hsiao *et al.*, 2022). Model 2 also incorporates year, country and industry-fixed effects to account for time-invariant industry differences and macroeconomic factors.

The analysis shows that female directors, board independence and CEOD significantly impact EI in European companies. Female directors and board independence have positive effects, whereas CEOD negatively influences EI. Board size, however, does not show a significant effect. As a result, hypotheses *H1*, *H3* and *H4* are supported, whereas *H2* is not.

4.3 Results discussion

Gender diversity on boards significantly enhances EI, as shown by a positive coefficient of 0.166 with a 1% p-value. This confirms the importance of female directors in promoting corporate strategies that focus on environmental priorities. Boards with higher female representation are more proactive in incorporating environmental issues into their agendas (Moreno-Ureba et al., 2022; Konadu et al., 2022). The European Commission's emphasis on gender equality aligns with these findings, as gender-diverse boards improve decisionmaking and strengthen stakeholder relations, enhancing a company's environmental and social performance (Farza et al., 2022; Nuber and Velte, 2021). Research suggests that gender diversity contributes unique perspectives that enhance innovation and competitiveness (Mirza et al., 2012). Empirical evidence also highlights that female directors' risk-averse and compliance-focused approaches positively affect employee engagement, creativity and overall firm progress (Nadeem et al., 2019). However, García-Meca et al. (2023) emphasized that female directors need sufficient authority on the board to maximize their influence on EI. Other studies indicate that gender-diverse boards are more effective in implementing EI strategies and improving CSR and environmental disclosures (Alia and Mardawi, 2021).

The theoretical foundations support these findings. According to Resource Dependency Theory, board diversity, mainly including female directors, improves firm performance by providing diverse knowledge, expertise and networks that reduce resource dependence (Borgelt and Falk, 2007). These directors' ethical and thoughtful approaches align closely with shareholders' interests, enhancing oversight of EI (Issa and Bensalem, 2023).

The analysis confirms the significance of board independence in improving EI among European companies. Independent boards enhance a company's reputation, ensure compliance with environmental regulations and secure legitimacy, which is essential for CG (Farza *et al.*, 2022). Independent directors mitigate conflicts between shareholders and management, aligning management actions with shareholders' interests and promoting long-term firm sustainability (Lu and Wang, 2018; Zhu *et al.*, 2022). From a stakeholder perspective, these directors maintain ethical standards and social responsibilities, which are vital for the firm's success (García-Sánchez *et al.*, 2020; Ntim and Soobaroyen, 2013). Resource Dependency Theory also highlights the role of independent directors in providing critical resources like expertise and connections, supporting innovation and enhancing environmental practices (Khaireddine *et al.*, 2020; Osemene *et al.*, 2021). Their presence fosters a culture of managerial risk-taking and innovation by enabling access to essential resources and opportunities (Lu and Wang, 2018; García-Sánchez *et al.*, 2020).

In addition, board independence benefits extend beyond shareholder and stakeholder interests. Independent directors protect their professional reputation, preventing misconduct and thus enhancing the firm's reputation (Ruiz-Castillo *et al.*, 2023). Their accountability toward stakeholders and the environment empowers them to positively influence EI (Nadeem *et al.*, 2020).

The regression analysis confirms *H4*, indicating that CEOD negatively impacts EI, as shown by a significant beta coefficient of –3.277 and a 1% *p*-value. This supports Agency Theory's argument that combining CEO and chair roles reduces board independence, impairing the board's capacity to promote environmental strategies (Krause *et al.*, 2013; Ghardallou, 2022). Separating these roles is essential for maintaining independence and unbiased decision-making, ensuring alignment with long-term shareholder and stakeholder interests, including environmental goals (Korir and Tenai, 2020; Rossi *et al.*, 2021). Studies further indicate that CEOD often prioritizes short-term financial performance at the expense

of social and environmental sustainability, hindering effective EI strategies (Zhang, 2012; Mallin and Michelon, 2011).

Research shows CEOD can limit board members' contributions, particularly female directors who positively influence ESG performance (Romano *et al.*, 2020; Ghardallou, 2022). Sustainability performance suffers under CEOD due to conflicts of interest and a lack of strategic focus on long-term environmental objectives (Rossi and Cebula, 2015). This emphasizes the importance of separating the roles to promote sustainable development.

The results indicate no significant impact of board size on EI, consistent with previous studies by Farza *et al.* (2022), Trisnawati *et al.* (2022) and Romano *et al.* (2020). The complexities of decision-making in larger boards may hinder their effectiveness in advancing EI (Cheng, 2008), as reduced motivation among members can dilute the board's effectiveness (Zona *et al.*, 2013).

The regression results show that all control variables significantly impact EI. SIZE and CSRCO positively correlate with EI, as larger firms have more resources to implement sustainable technologies (Zhu *et al.*, 2022; Dwekat *et al.*, 2022a). Larger organizations can manage the substantial investments required for EI, leveraging their capabilities for strategic advantage. The presence of CSR committees further supports sustainability efforts by integrating environmental considerations into business strategies (Dwekat *et al.*, 2022a).

Leverage (LEV), however, negatively affects EI, indicating that high debt levels can constrain a company's ability to fund environmental projects. Firms with high leverage may prioritize financial obligations over long-term sustainability investments, potentially limiting their EI activities (Nadeem *et al.*, 2020). Conversely, firms with high ROE are more likely to support EI initiatives, as they have the financial capacity to invest in sustainable projects, enhancing long-term environmental performance. This relationship between financial health and EI demonstrates that successful firms are better positioned to integrate sustainability into their core operations (Nadeem *et al.*, 2020).

#### 4.4 The moderating impact of CSR committee

The results in Table 5 highlight the moderating role of the CSRC on the relationship between board characteristics (gender diversity, board size, board independence and CEOD) and EI, supporting *H*5. The findings indicate that CSRC significantly enhances these associations, demonstrating its positive influence across various board dimensions. Notably, while the direct impact of CEOD on EI was negative, suggesting that the dual roles could hinder environmental efforts due to conflicts of interest or power concentration (as shown in Table 4), the presence of CSRC effectively mitigates these risks. CSRC provides oversight and strategic alignment with environmental goals, counteracting the governance issues linked to CEOD (Moreno-Ureba *et al.*, 2022).

The CSRC's ability to moderate these effects underscores its crucial role in guiding environmental decisions and strategies, strengthening the board's capacity to address complex environmental and social issues (Moreno-Ureba *et al.*, 2022; Dixon-Fowler *et al.*, 2017). Empirical evidence further supports the impact of CSRCs in aligning board characteristics with CSR outcomes, thereby enhancing sustainability practices (Pucheta-Martínez and Gallego-Álvarez, 2019; Endrikat *et al.*, 2021). These findings highlight the CSRC's essential role in not only supporting but also enhancing the effectiveness of EI efforts led by diverse and structured boards.

#### 4.5 Robustness tests

4.5.1 *Generalized estimating equation*. In our research, we implemented the Generalized Estimating Equation (GEE) approach, consistent with recent studies (Bolourian *et al.*, 2023).

**Table 5.** The moderate role of CSR committee

WADIADIEC	(1)
VARIABLES	Innovation
CSRC	30.57***(9.395)
BOGE	0.0481**(0.0749)
CSRC*BOGE	0.170**(0.0829)
BOSI	3.493*(3.122)
CSRC*BOSI	12.03***(3.526)
BOIN	0.0139**(0.0370)
CSRC*BOIN	0.119***(0.0446)
CEOD	-8.650***(2.460)
CSRC*CEOD	6.006**(2.726)
SIZE	6.213***(0.465)
LEC	-35.35***(3.741)
ROE	0.0146**(0.00588)
Constant	-41.52***(11.11)
Year fixed effects	Yes
Country fixed effects	Yes
Industry fixed effects	Yes
Observations	6,126
Adjusted R-squared	0.261

**Notes:** The table examines the moderating effect of the CSRC on the relationship between board characteristics and EI. To ensure analytical precision, robust standard errors, adjusted for heteroscedasticity and clustered at the firm and industry levels, are applied. The regression models incorporate controls for industry, year and country to account for time-invariant industry differences and macroeconomic variations. Detailed variable definitions are available in Table 1. Statistical significance is denoted as follows: \*\*\* for 1%, \*\* for 5% and \* for 10%

Source: Authors' own work

This method is ideally suitable for data that may exhibit dependencies, which could lead to autocorrelation issues. It offers a more dependable and robust framework than traditional random and fixed effects models (Liang and Zeger, 1986). The results from Model 2 in Table 6 confirm the robustness of our findings, showing that the impact of board characteristics on EI is consistent with the main model's results in Table 4.

4.5.2 Control for the endogeneity. Endogeneity is a critical concern in governance research, particularly when examining the influence of board characteristics on EI, as director appointments may be biased by a firm's operational and informational context (Nadeem *et al.*, 2020). To address this, the study uses lagged models, positioning independent variables one period ahead of the dependent variables to establish causality. Using this approach, the results from Model 1 in Table 6 are consistent with the main findings, reinforcing the causal relationship.

Furthermore, in line with Nadeem *et al.* (2020), the study applies the two-stage least squares (2SLS) method to control for endogeneity. This technique uses instruments correlated with the endogenous predictors but uncorrelated with the error terms, enabling a clearer interpretation of effects. The results from Model 3 in Table 6, which implements 2SLS, confirm the robustness of the study's findings, demonstrating that the methods effectively manage endogeneity issues.

4.5.3 Control for CSR-sensitive industries. To control for CSR-sensitive industries in examining the impact of board characteristics on EI, the study focuses on sectors such as

Table 6. Different model specifications

VARIABLES	(1)	(2)	(3)
	One-year lag	GEE	2SLS
	Innovation	Innovation	Innovation
BOGE	0.160***(0.0460)	0.0783(0.0477)	0.158***(0.0545)
BOSI	6.301*(1.863)	5.024*(1.906)	6.651(2.212)
BOIN	0.106***(0.0241)	0.0668***(0.0224)	0.118***(0.0272)
CEOD	-3.443***(1.170)	-3.713***(1.295)	-1.650**(1.353)
SIZE	6.452***(0.480)	6.529***(0.461)	7.029***(0.492)
LEV	-40.89***(3.991)	-36.04***(3.721)	-23.96***(3.801)
ROE	0.0103*(0.00589)	0.0157*(0.00921)	0.00279(0.00641)
CSRC	8.220***(1.278)	11.10***(1.210)	8.222***(1.318)
Constant	-85.65***(7.014)	-75.63***(7.824)	-96.97***(6.591)
Year fixed effects	Yes	Yes	Yes
Country fixed effects Industry fixed effects	Yes	Yes	Yes
	Yes	Yes	Yes
Observations	5,297	6,126	5,601
Adjusted <i>R</i> -squared	0.220		0.151

**Notes:** The table presents the effects of board characteristics on EI using various regression methods. Model 1 applies a one-year lag for the independent variables, Model 2 uses the GEE model and Model 3 uses the 2SLS method. Robust standard errors, adjusted for heteroscedasticity and clustered at the firm and industry levels, are used to improve accuracy. The models incorporate industry-, year- and country-specific controls to account for time-invariant industry differences and macroeconomic variations. Details on the variables used are available in Table 1. Statistical significance levels are indicated as follows: \*\*\* for 1%, \*\* for 5% and \* for 10%

Source: Authors' own work

utilities, mining and production – industries with substantial environmental footprints and a strong incentive to maintain a positive social image (Al-Shaer and Zaman, 2018; Meqbel *et al.*, 2024). These industries comprise around 20% of the firm-year observations in the data set. Consistent with previous research (Dwekat *et al.*, 2022a; Meqbel *et al.*, 2024), the results in Table 7 reveal a significant positive relationship between CSR-sensitive industries and EI, suggesting that firms in these sectors are more inclined to adopt innovative environmental practices. The coefficients and significance levels of the main variables remain consistent with those presented in Table 4, reinforcing the robustness of the findings.

4.5.4 Control for country-level effects. To account for country-level influences on environmental practices, such as legal frameworks, sociocultural factors and cultural dimensions (Meqbel et al., 2024; Dwekat et al., 2022a), this analysis replaces country-specific indicators with broader measures like regulatory quality and cultural dimensions (e. g. power distance, individualism, masculinity and uncertainty avoidance). Regulatory quality, indicating the government's effectiveness in implementing sustainable regulations (Mooneeapen et al., 2022), is associated with improved sustainability performance (Meqbel et al., 2024; Rahi et al., 2023), as companies in such environments are more likely to adopt responsible practices (Uyar et al., 2020). Cultural dimensions also shape CSR practices by influencing stakeholder orientations and social priorities (Orij, 2010). Khlif et al. (2015) suggested that leadership styles, resource strategies and ethical decision-making linked to national culture impact sustainability efforts. The empirical results, as shown in Model 1 in Table 7, confirm a significant positive relationship between board characteristics and EI,

Table 7. Control for country-level factors and CSR-sensitive industries

VADIADIEC	(1) Innovation	(2) Innovation
VARIABLES	IIIIOVation	IIIIOVatioii
BOGE	0.0961**(0.0455)	0.0987**(0.0490)
BOSI	5.725*(1.811)	3.806*(1.964)
BOIN	0.0897***(0.0219)	0.0438*(0.0233)
CEOS	-3.445***(1.196)	-1.135(1.282)
SIZE	6.463***(0.449)	6.822***(0.467)
LEV	-37.49***(3.687)	-25.01***(3.628)
ROE	0.0168***(0.00606)	0.0200***(0.00613)
CSRC	10.83***(1.204)	11.14***(1.218)
Power distance	-0.307***(0.0863)	
Individualism	-0.118(0.0843)	
Masculinity	-0.152***(0.0303)	
Uncertainty avoidance	0.192***(0.0716)	
Rule of law	1.072(1.514)	
CSR_sensitive_industry		6.043***(0.972)
Constant	-72.50***(9.560)	-76.01***(8.350)
Year fixed effects	Yes	Yes
Country fixed effects	No	Yes
Industry fixed effects	Yes	No
Observations	6,109	6,126
Adjusted R-squared	0.251	0.204

**Notes:** The table illustrates the impact of board characteristics on EI while controlling for country-specific factors and CSR-sensitive industries. Model 1 incorporates cultural factors, whereas Model 2 controls for CSR-sensitive industries. To ensure precision, robust standard errors, adjusted for heteroscedasticity and clustered at the firm and industry levels, are used. The models also account for industry-, year- and country-specific controls to address time-invariant industry differences and macroeconomic fluctuations. Detailed variable information is provided in Table 1. Statistical significance is denoted as follows: \*\*\* for 1%, \*\* for 5% and \* for 10%

Source: Authors' own work

even after accounting for these country-level factors, supporting the robustness of our findings.

#### 5. Conclusion

This study addresses a significant gap in the literature by analyzing the impact of board characteristics – gender diversity, independence, size and CEOD – on EI among European firms listed on the STOXX Europe index from 2011 to 2022, using a data set of 6,126 firm-year observations. It also investigates the moderating role of CSR committees in these relationships.

The findings emphasize the importance of gender diversity and board independence in promoting EI, supported by legitimacy, agency and stakeholder theories. Female board members and independent directors are linked to stronger environmental commitments. In contrast, CEOD negatively impacts EI, aligning with the notion that concentrated power hinders the board's ability to promote sustainable practices. No significant effect of board size on EI was found. Control variables, such as CSR committees and firm size, positively influence EI, with larger companies and those with CSR committees more committed to sustainability efforts. Financial leverage negatively impacts EI, whereas profitability (ROE) has a positive effect.

The study highlights the significant moderating role of CSR committees, which enhance the relationship between board characteristics and EI. CSR committees effectively mitigate the negative impact of CEOD by promoting oversight and strategic alignment with environmental goals, thereby improving corporate sustainability practices.

These findings have important theoretical and practical implications. The positive impact of gender diversity underscores the need for greater female representation on boards to bring unique perspectives and drive environmental initiatives. In addition, the importance of board independence suggests that companies should prioritize appointing knowledgeable, independent directors to integrate sustainability into corporate strategies. Regulatory support for board independence is essential for effective governance and proactive engagement in EI.

The negative effect of CEOD supports separating the roles of CEO and board chair to avoid power concentration and maintain unbiased, long-term decision-making. Establishing CSR committees further strengthens the effectiveness of board characteristics in promoting EI, ensuring rigorous oversight and strategic alignment with environmental objectives.

Theoretically, this study advances the understanding of CG and EI by demonstrating how specific board characteristics – gender diversity, board independence and CEOD – impact environmental strategies. It supports the application of legitimacy, agency and stakeholder theories, showing that diverse and independent boards enhance a firm's legitimacy, reduce conflicts of interest and address environmental concerns more effectively.

The study provides a foundation for future research to explore the causal mechanisms and other moderating factors affecting these relationships. Further investigation into board member qualifications and experiences could deepen the understanding of their impact on environmental strategies. In addition, future studies could examine board characteristics in developing countries, considering the influence of cultural factors on EI (Ullah *et al.*, 2022).

#### References

- Abdelhaq, R. and Dwekat, A. (2024), "Board committees and voluntary disclosure: evidence from Palestine", *Artificial Intelligence and Economic Sustainability in the Era of Industrial Revolution* 5.0, Springer Nature Switzerland, Cham, pp. 25-38.
- Abdelhaq, R., Dwekat, A., Atout, S. and Nour, A.I. (2024), "The impact of board characteristics on the level of voluntary disclosure: evidence from Palestinian listed companies", *Artificial Intelligence and Economic Sustainability in the Era of Industrial Revolution 5.0*, Springer Nature Switzerland, Cham, pp. 107-122.
- Abweny, M., Afrifa, G.A. and Iqbal, A. (2024), "The complementarity and substitution effects of CSR-focused governance mechanisms on CSR decoupling", *Corporate Governance: An International Review*,
- Agustia, D. (2023), "Does green innovation play an important role in the effect board gender diversity has on firm performance?", *Intangible Capital*, Vol. 19 No. 2, pp. 146-164.
- Ahmed, R., Abweny, M., Benjasak, C. and Nguyen, D.T. (2024), "Financial sanctions and environmental, social, and governance (ESG) performance: a comparative study of ownership responses in the Chinese context", *Journal of Environmental Management*, Vol. 351, p. 119718.
- Albitar, K. and Hussainey, K. (2023), "Sustainability, environmental responsibility and innovation", *Green Finance*, Vol. 5 No. 1, pp. 85-88.
- Albitar, K., Nasrallah, N., Hussainey, K. and Wang, Y. (2024), "Eco-innovation and corporate waste management: the moderating role of ESG performance", *Review of Quantitative Finance and Accounting*, Vol. 63 No. 2, pp. 1-25.
- Alia, M.A. and Mardawi, Z. (2021), "The impact of ownership structure and board characteristics on corporate social responsibility disclosed by Palestinian companies", *Jordan Journal of Business Administration*, Vol. 17 No. 2.

- Journal of Financial Reporting and Accounting
- Alia, M.A., Dwekat, A., Ismail, T., Al-Saber, D. and Salman, L. (2024), "CSRD in the Arab world: the role of audit quality", *Artificial Intelligence and Economic Sustainability in the Era of Industrial Revolution 5.0*, Springer Nature Switzerland, Cham, pp. 123-135.
- Alkaraan, F. (2022), "A new era of mergers and acquisitions: towards synergy between industry 4.0 and circular economy", *Advances in Mergers and Acquisitions*, Vol. 21, pp. 51-61, doi: 10.1108/S1479-361X20220000021004.
- Alkaraan, F., Elmarzouky, M., Hussainey, K., Venkatesh, V.G., Shi, Y. and Gulko, N. (2024), "Reinforcing green business strategies with Industry 4.0 and governance towards sustainability: Natural-resource-based view and dynamic capability", *Business Strategy and the Environment*, Vol. 33 No. 4, pp. 1-19, doi: 10.1002/bse.3665.
- Almaqtari, F.A., Elsheikh, T., Al-Hattami, H.M. and Mishra, N. (2023), "The impact of board characteristics on environmentally friendly production: a cross country study in Asia and Europe", *Journal of Cleaner Production*, Vol. 392, p. 136257.
- Alqatan, A. (2024), "The impact of board diversity on earnings management in Kuwait", *Journal of Financial Reporting and Accounting*, doi: 10.1108/JFRA-05-2024-0279.
- Al-Shaer, H. and Zaman, M. (2018), "Credibility of sustainability reports: the contribution of audit committees", *Business Strategy and the Environment*, Vol. 27 No. 7, pp. 973-986, doi: 10.1002/bse.2046.
- Alta'any, M., Kayed, S., Meqbel, R. and Albitar, K. (2024), "Speaking success: managerial tone in earnings conference calls and financial performance", *Corporate Governance: The International Journal of Business in Society.*
- Baraibar-Diez, E. and Odriozola, M. (2019), "CSR committees and their effect on ESG performance in UK, France, Germany, and Spain", *Sustainability*, Vol. 11 No. 18, p. 5077.
- Belhouchet, S. and Chouaibi, J. (2024), "The impact of audit committee attributes on integrated reporting quality: evidence from European companies listed on the STOXX Europe 600 index", *Meditari Accountancy Research*, Vol. 32 No. 4.
- Bolourian, S., Alinaghian, L. and Angus, A. (2023), "Exploring the role of board-level corporate social responsibility committees in corporate social responsibility performance: a configurational approach", *Journal of Business Research*, Vol. 169, p. 114280.
- Borgelt, K. and Falk, I. (2007), "The leadership/management conundrum: innovation or risk management?", *Leadership and Organization Development Journal*, Vol. 28 No. 2, pp. 122-136.
- Brunnermeier, S.B. and Cohen, M.A. (2003), "Determinants of environmental innovation in US manufacturing industries", *Journal of Environmental Economics and Management*, Vol. 45 No. 2, pp. 278-293.
- Carmona, P., Dwekat, A. and Mardawi, Z. (2022), "No more black boxes! Explaining the predictions of a machine learning XGBoost classifier algorithm in business failure", *Research in International Business and Finance*, Vol. 61, p. 101649.
- Cheng, S. (2008), "Board size and the variability of corporate performance", *Journal of Financial Economics*, Vol. 87 No. 1, pp. 157-176, doi: 10.1016/j.jfineco.2006.10.006.
- Chouaibi, J. and Jarboui, A. (2012), "Investigating the effect of the directors' board characteristics on financial performance through innovation activities", *International Journal of Behavioural Accounting and Finance*, Vol. 3 Nos 1/2, pp. 65-87.
- Del Río, P., Peñasco, C. and Romero-Jordán, D. (2016), "What drives eco-innovators? A critical review of the empirical literature based on econometric methods", *Journal of Cleaner Production*, Vol. 112, pp. 2158-2170.
- Dhaliwal, D.S., Li, O.Z., Tsang, A. and Yang, Y.G. (2011), "Voluntary nonfinancial disclosure and the cost of equity capital: the initiation of corporate social responsibility reporting", *The Accounting Review*, Vol. 86 No. 1, pp. 59-100.
- Dixon-Fowler, H.R., Ellstrand, A.E. and Johnson, J.L. (2017), "The role of board environmental committees in corporate environmental performance", *Journal of Business Ethics*, Vol. 140 No. 3, pp. 423-438.

- Dwekat, A., Seguí-Mas, E. and Tormo-Carbó, G. (2020b), "The effect of the board on corporate social responsibility: bibliometric and social network analysis", *Economic Research-Ekonomska Istraživanja*, Vol. 33 No. 1, pp. 3580-3603.
- Dwekat, A., Meqbel, R., Seguí-Mas, E. and Tormo-Carbó, G. (2022a), "The role of the audit committee in enhancing the credibility of CSR disclosure: evidence from STOXX Europe 600 members", *Business Ethics A European Review*, doi: 10.1111/beer.12439.
- Dwekat, A., Seguí-Mas, E., Zaid, M.A. and Tormo-Carbó, G. (2022b), "Corporate governance and corporate social responsibility: mapping the most critical drivers in the board academic literature", *Meditari Accountancy Research*, Vol. 30 No. 6, pp. 1705-1739.
- Eberhardt-Toth, E. (2017), "Who should be on a board corporate social responsibility committee?", *Journal of Cleaner Production*, Vol. 140, pp. 1926-1935.
- Elad Fotoh, L., Ngan Wong, M. and Solange Bongbee, N. (2018), "The role of board characteristics in effective corporate governance: the case of airbus group)", *International Journal of Social Sciences Perspectives*, Vol. 2 No. 1, pp. 87-95, doi: 10.33094/7.2017.2018.21.87.95.
- Endrikat, J., De Villiers, C., Guenther, T.W. and Guenther, E.M. (2021), "Board characteristics and corporate social responsibility: a meta-analytic investigation", *Business and Society*, Vol. 60 No. 8, pp. 2099-2135.
- Estapé-Dubreuil, G., Ashta, A. and Hédou, J.-P. (2016), "Micro-equity for sustainable development: Selection, monitoring and exit strategies of micro-angels", *Ecological Economics*, Vol. 130, pp. 117-129, doi: 10.1016/j.ecolecon.2016.06.021.
- European Commission (2021), "2021 Strategic foresight report: the EU's capacity and freedom to act", Publications Office of the European Union, available at: https://ec.europa.eu
- Farza, K., Ftiti, Z., Hlioui, Z., Louhichi, W. and Omri, A. (2022), "The effect of corporate board characteristics on environmental innovation", *Environmental Modeling and Assessment*, Vol. 27 No. 6, pp. 1021-1042, available at: https://link.springer.com/article/10.1007/s10666-022-09836-3
- Freeman, R.E. and Evan, W.M. (1990), "Corporate governance: a stakeholder interpretation", *Journal of Behavioral Economics*, Vol. 19 No. 4, pp. 337-359.
- Freeman, R.E., Harrison, J.S., Wicks, A.C., Parmar, B.L. and De Colle, S. (2010), "Stakeholder theory: the state of the art".
- Fuente, J.A., García-Sánchez, I.M. and Lozano, M.B. (2017), "The role of the board of directors in the adoption of GRI guidelines for the disclosure of CSR information", *Journal of Cleaner Production*, Vol. 141, pp. 737-750, doi: 10.1016/j.jclepro.2016.09.155.
- Galia, F., Zenou, E. and Ingham, M. (2015), "Board composition and environmental innovation: does gender diversity matter?", *International Journal of Entrepreneurship and Small Business*, Vol. 24 No. 1, pp. 117-141.
- García Martín, C.J. and Herrero, B. (2020), "Do board characteristics affect environmental performance? A study of EU firms", *Corporate Social Responsibility and Environmental Management*, Vol. 27 No. 1, pp. 74-94.
- García-Meca, E., Ramón-Llorens, M.C. and Martínez-Ferrero, J. (2023), "Feminine expertise on board and environmental innovation: the role of critical mass", *Review of Managerial Science*, Vol. 18 No. 8, pp. 1-32.
- García-Sánchez, I.-M., Aibar-Guzmán, C. and Aibar-Guzmán, B. (2020), "The effect of institutional ownership and ownership dispersion on eco-innovation", *Technological Forecasting and Social Change*, Vol. 158, pp. 120-173, doi: 10.1016/j.techfore.2020.120173.
- García-Sánchez, I.M., Gómez-Miranda, M.E., David, F. and Rodríguez-Ariza, L. (2019a), "Board independence and GRI-IFC performance standards: the mediating effect of the CSR committee", *Journal of Cleaner Production*, Vol. 225, pp. 554-562.

- García-Sánchez, I.-M.G.-S., Hussain, N. and Martínez-Ferrero, J. (2019b), "An empirical analysis of the complementarities and substitutions between effects of CEO ability and corporate governance on socially responsible performance", *Journal of Cleaner Production*, Vol. 215, pp. 1288-1300, doi: 10.1016/j.jclepro.2019.01.130.
- García-Sánchez, I.M., Gallego-Álvarez, I. and Zafra-Gómez, J.L. (2021), "Do independent, female and specialist directors promote eco-innovation and eco-design in agri-food firms?", *Business Strategy and the Environment*, Vol. 30 No. 2, pp. 1136-1152.
- Gennari, F. (2019), "How to lead the board of directors to a sustainable development of business with the CSR committees", *Sustainability*, Vol. 11 No. 24, p. 6987.
- Gennari, F. and Salvioni, D.M. (2019), "CSR committees on boards: the impact of the external country level factors". *Journal of Management and Governance*, Vol. 23 No. 3, pp. 759-785.
- Ghardallou, W. (2022), "Corporate sustainability and firm performance: the moderating role of CEO education and tenure", *Sustainability*, Vol. 14 No. 6, p. 3513, doi: 10.3390/su14063513.
- Ginesti, G., Campa, D., Allini, A. and Maffei, M. (2023), "The role of CSR committee characteristics on RandD investments", *International Business Review*, Vol. 32 No. 5, p. 102147.
- Godos-Díez, J.-L., Cabeza-García, L., Alonso-Martínez, D. and Fernández-Gago, R. (2018), "Factors influencing board of directors' decision-making process as determinants of CSR engagement", *Review of Managerial Science*, Vol. 12 No. 1, pp. 229-253, doi: 10.1007/s11846-016-0220-1.
- Goodstein, J., Gautam, K. and Boeker, W. (1994), "The effects of board size and diversity on strategic change", *Strategic Management Journal*, Vol. 15 No. 3, pp. 241-250.
- Hillman, A.J., Withers, M.C. and Collins, B.J. (2009), "Resource dependence theory: a review", *Journal of Management*, Vol. 35 No. 6, pp. 1404-1427.
- Hsiao, C., Shi, Z. and Zhou, Q. (2022), "Transformed estimation for panel interactive effects models", *Journal of Business and Economic Statistics*, Vol. 40 No. 4, pp. 1831-1848.
- Hunjra, A.I. and Hussainey, K. (Eds) (2024), "The Emerald Handbook of ethical finance and corporate social responsibility: a framework for sustainable development".
- Hussainey, K., Albitar, K. and Alkaraan, F. (2022), "Corporate narrative reporting on Industry 4.0 technologies: does governance matter?", *International Journal of Accounting and Information Management*, Vol. 30 No. 4, pp. 1834-7649, doi: 10.1108/IJAIM-02-2022-0024.
- Issa, A. and Bensalem, N. (2023), "Are gender-diverse boards eco-innovative? The mediating role of corporate social responsibility strategy", *Corporate Social Responsibility and Environmental Management*, Vol. 30 No. 2, pp. 742-754.
- Javeed, S.A., Latief, R., Cai, X., San Ong, T., Qian, S. and Haq, A.U. (2022), "What is the role of the board sustainable committee for corporate social responsibility? The moderating effect of gender diversity and ownership concentration", *Journal of Cleaner Production*, Vol. 379, p. 134710.
- Jensen, M.C. (1993), "The modern industrial revolution, exit, and the failure of internal control systems", *The Journal of Finance*, Vol. 48 No. 3, pp. 831-880.
- Kayed, S. and Meqbel, R. (2024), "Earnings management and tone management: evidence from FTSE 350 companies", *Journal of Financial Reporting and Accounting*, Vol. 22 No. 4, pp. 842-867.
- Kayed, S., Alta'any, M., Meqbel, R., Khatatbeh, IN. and Mahafzah, A. (2024), "Bank FinTech and bank performance: evidence from an emerging market", *Journal of Financial Reporting and Accounting*.
- Khaireddine, H., Salhi, B., Aljabr, J. and Jarboui, A. (2020), "Impact of board characteristics on governance, environmental and ethical disclosure", *Society and Business Review*, Vol. 15 No. 3, pp. 273-295.
- Khan, S.A.R., Razzaq, A., Yu, Z. and Miller, S. (2021), "Retracted: Industry 4.0 and circular economy practices: a new era business strategies for environmental sustainability", *Business Strategy and the Environment*, Vol. 30 No. 8, pp. 4001-4014.

- Khlif, H., Guidara, A. and Souissi, M. (2015), "Corporate social and environmental disclosure and corporate performance: evidence from South Africa and Morocco", *Journal of Accounting in Emerging Economies*, Vol. 5 No. 1, pp. 51-69.
- Konadu, R., Ahinful, G.S., Boakye, D.J. and Elbardan, H. (2022), "Board gender diversity, environmental innovation and corporate carbon emissions", *Technological Forecasting and Social Change*, Vol. 174, p. 121279.
- Korir, F.J. and Tenai, J.K. (2020), "Does structural power matter? Board attributes and firm performance: moderated by CEO duality", *SEISENSE Journal of Management*, Vol. 3 No. 5, pp. 54-64, available at: https://journal.seisense.com/index.php/jom/article/view/432
- Krause, R., Semadeni, M. and Cannella, A.A. (2013), "CEO duality: a review and research agenda", Journal of Management, Vol. 40 No. 1, doi: 10.1177/0149206313503013.
- Lacetera, N. (2001), "Corporate governance and the governance of innovation: the case of pharmaceutical industry", *Journal of Management and Governance*, Vol. 5 No. 1, pp. 29-59.
- Liang, K.Y. and Zeger, S.L. (1986), "Longitudinal data analysis using generalized linear models", *Biometrika*, Vol. 73 No. 1, pp. 13-22.
- Lu, J. and Wang, W. (2018), "Managerial conservatism, board independence and corporate innovation", *Journal of Corporate Finance*, Vol. 48, pp. 1-16, doi: 10.1016/j.jcorpfin.2017.10.016.
- Mallin, C.A. and Michelon, G. (2011), "Board reputation attributes and corporate social performance: an empirical investigation of the US best corporate citizens", *Accounting and Business Research*, Vol. 41 No. 2, pp. 119-144, doi: 10.1080/00014788.2011.550740.
- Mardawi, Z., Dwekat, A., Meqbel, R. and Ibáñez, P.C. (2023), "Configurational analysis of corporate governance and corporate social responsibility reporting assurance: understanding the role of board and CSR committee", *Meditari Accountancy Research*, Vol. 32 No. 2.
- Meckling, W.H. and Jensen, M.C. (1976), "Theory of the firm", *Managerial Behavior*, *Agency Costs and Ownership Structure*, Springer, Cham.
- Meqbel, R., Alta'any, M., Kayed, S. and Al-Omush, A. (2024), "Earnings management and sustainability assurance: the moderating role of CSR committee", *Corporate Social Responsibility and Environmental Management*, Vol. 31 No. 3, pp. 1769-1785.
- Miller, T. and del Carmen Triana, M. (2009), "Demographic diversity in the boardroom: Mediators of the board diversity—firm performance relationship", *Journal of Management Studies*, Vol. 46 No. 5, pp. 755-786.
- Mirza, H.H., Mahmood, S., Andleeb, S. and Ramzan, F. (2012), "Gender diversity and firm performance: evidence from Pakistan", *Journal of Social and Development Sciences*, Vol. 3 No. 5, pp. 161-166, available at: https://core.ac.uk/download/pdf/288023449.pdf
- Mohamed Riyath, M.I. and Inun Jariya, A.M. (2024), "The role of ESG reporting, artificial intelligence, stakeholders and innovation performance in fostering sustainability culture and climate resilience", *Journal of Financial Reporting and Accounting*, doi: 10.1108/JFRA-10-2023-0621.
- Mongo, M., Belaïd, F. and Ramdani, B. (2021), "The effects of environmental innovations on CO2 emissions: Empirical evidence from Europe", Environmental Science and Policy, Vol. 118, pp. 1-9.
- Mooneeapen, O., Abhayawansa, S. and Mamode Khan, N. (2022), "The influence of the country governance environment on corporate environmental, social and governance (ESG) performance", *Sustainability Accounting, Management and Policy Journal*, Vol. 13 No. 4, pp. 953-985.
- Moreno-Gómez, J., Lagos, D. and Gómez-Betancourt, G. (2017), "Effect of the board of directors on firm performance", *International Journal of Economic Research*, Vol. 14 No. 6, pp. 349-361. www. researchgate.net/publication/314156794\_Effect\_of\_the\_Board\_of\_Directors\_on\_firm\_performance
- Moreno-Ureba, E., Bravo-Urquiza, F. and Reguera-Alvarado, N. (2022), "An analysis of the influence of female directors on environmental innovation: when are women greener?", *Journal of Cleaner Production*, Vol. 374, available at: www.sciencedirect.com/science/article/pii/S095965262203445X

- Journal of Financial Reporting and Accounting
- Nadeem, M., Suleman, T. and Ahmed, A. (2019), "Women on boards, firm risk and the profitability nexus: does gender diversity moderate the risk and return relationship?", *International Review of Economics and Finance*, Vol. 64, pp. 427-442, doi: 10.1016/j.iref.2019.08.007.
- Nadeem, M., Bahadar, S., Gull, A.A. and Iqbal, U. (2020), "Are women eco-friendly? Board gender diversity and environmental innovation", *Business Strategy and the Environment*, Vol. 29 No. 8, pp. 3146-3161, doi: 10.1002/bse.2563.
- Nandy, M., Hussainey, K. and Lodh, S. (2024), "Social impact of social responsibility of business", International Journal of Business Governance and Ethics.
- Ntim, C. and Soobaroyen, T. (2013), "Corporate governance and performance in socially responsible corporations: new empirical insights from a neo-institutional framework", *Corporate Governance: An International Review*, Vol. 21 No. 5, pp. 468-494, doi: 10.1111/ corg.12026.
- Nuber, C. and Velte, P. (2021), "Board gender diversity and carbon emissions: European evidence on curvilinear relationships and critical mass", *Business Strategy and the Environment*, Vol. 30 No. 4, pp. 1958-1992, doi: 10.1002/bse.2727.
- Omran, M.S., Zaid, M.A. and Dwekat, A. (2021), "The relationship between integrated reporting and corporate environmental performance: a green trial", *Corporate Social Responsibility and Environmental Management*, Vol. 28 No. 1, pp. 427-445.
- Orazalin, N. (2020), "Do board sustainability committees contribute to corporate environmental and social performance? The mediating role of corporate social responsibility strategy", *Business Strategy and the Environment*, Vol. 29 No. 1, pp. 140-153.
- Orij, R. (2010), "Corporate social disclosures in the context of national cultures and stakeholder theory", *Accounting, Auditing and Accountability Journal*, Vol. 23 No. 7, pp. 868-889.
- Osemene, O.F., Adinnu, P., Fagbemi, T.O. and Olowookere, J.K. (2021), "Corporate governance and environmental accounting reporting in selected quoted African companies", *Global Business Review*. Vol. 25 No. 4.
- Pucheta-Martínez, M.C. and Gallego-Álvarez, I. (2019), "An international approach of the relationship between board attributes and the disclosure of corporate social responsibility issues", *Corporate Social Responsibility and Environmental Management*, Vol. 26 No. 3, pp. 612-627.
- Qaderi, S.A., Ghaleb, B.A.A., Hashed, A.A., Chandren, S. and Abdullah, Z. (2022), "Board characteristics and integrated reporting strategy: does sustainability committee matter?", Sustainability, Vol. 14 No. 10, p. 6092.
- Radu, C. and Smaili, N. (2021), "Alignment versus monitoring: an examination of the effect of the CSR committee and CSR-linked executive compensation on CSR performance", *Journal of Business Ethics*, Vol. 180 No. 1, pp. 1-19.
- Rahi, A.F., Chowdhury, M.A.F., Johansson, J. and Blomkvist, M. (2023), "Nexus between institutional quality and corporate sustainable performance: European evidence", *Journal of Cleaner Production*, Vol. 382, p. 135188.
- Rahman, H.U., Zahid, M., Jan, A., Al-Faryan, M.A.S. and Hussainey, K. (2024), "Is it the mere female directors or their attributes that matter for the quality of corporate sustainability disclosures?", *Business Strategy and the Environment*, Vol. 33 No. 2, pp. 661-678.
- Ramdhony, D., Omran, M. and Hussainey, K. (2023), "Board attributes and corporate social responsibility disclosure quality based on information typology", *Review of Accounting and Finance*, Vol. 22 No. 4, pp. 508-531.
- Romano, M., Cirillo, A., Favino, C. and Netti, A. (2020), "ESG (environmental, social and governance) performance and board gender diversity: the moderating role of CEO duality", *Sustainability*, Vol. 12 No. 21, pp. 92-98, doi: 10.3390/su12219298.

- Rossi, F. and Cebula, R.J. (2015), "Ownership structure and RandD: an empirical analysis of Italian listed companies", *PSL Quarterly Review*, Vol. 68 No. 275.
- Rossi, M., Chouaibi, J., Chouaibi, S., Jilani, W. and Chouaibi, Y. (2021), "Does a board characteristic moderate the relationship between CSR practices and financial performance? Evidence from European ESG firms", *Journal of Risk and Financial Management*, Vol. 14 No. 8, p. 354, doi: 10.3390/jrfm14080354.
- Ruiz-Castillo, M., Aragón-Correa, J.A. and Hurtado-Torres, N.E. (2023), "Independent directors and environmental innovations: how the visibility of public and private shareholders' environmental activism moderates the influence of board independence", *Business Strategy and the Environment*, Vol. 33 No. 2, p. 3498.
- Shapiro, D., Tang, Y., Wang, M. and Zhang, W. (2015), "The effects of corporate governance and ownership on the innovation performance of Chinese SMEs", *Journal of Chinese Economic and Business Studies*, Vol. 13 No. 4, pp. 311-335.
- Song, M., Wang, S. and Zhang, H. (2020), "Could environmental regulation and RandD tax incentives affect green product innovation?", *Journal of Cleaner Production*, Vol. 258, p. 120849.
- Spitzeck, H. (2009), "The development of governance structures for corporate responsibility", *Corporate Governance: The International Journal of Business in Society*, Vol. 9 No. 4, pp. 495-505.
- Traversi, M., Bannò, M. and Filippi, E. (2024), "The role of the cultural context in moderating the effect between the presence of women directors and firm environmental innovation", *Corporate Social Responsibility and Environmental Management*, Vol. 31 No. 5.
- Trisnawati, R., Wardati, S.D. and Putri, E. (2022), "The influence of majority ownership, profitability, size of the board of directors, and frequency of board of commissioners meetings on sustainability report disclosure", *Riset Akuntansi Dan Keuangan Indonesia*, Vol. 7 No. 1, pp. 94-104.
- Ullah, S., Agyei-Boapeah, H., Kim, J.R. and Nasim, A. (2022), "Does national culture matter for environmental innovation? A study of emerging economies", *Technological Forecasting and Social Change*, Vol. 181, p. 121755.
- Uyar, A., Kilic, M., Koseoglu, M.A., Kuzey, C. and Karaman, A.S. (2020), "The link among board characteristics, corporate social responsibility performance, and financial performance: evidence from the hospitality and tourism industry", *Tourism Management Perspectives*, Vol. 35, p. 100714.
- Velte, P. (2024), "Sustainable board governance and sustainable supply chain reporting: European evidence", *Journal of Strategy and Management*, Vol. 17 No. 2, pp. 222-243.
- Velte, P. and Stawinoga, M. (2020), "Do chief sustainability officers and CSR committees influence CSR-related outcomes? A structured literature review based on empirical-quantitative research findings", *Journal of Management Control*, Vol. 31 No. 4, pp. 333-377.
- Voinea, C.L., Rauf, F., Naveed, K. and Fratostiteanu, C. (2022), "The impact of CEO duality and financial performance on CSR disclosure: Empirical evidence from state-owned enterprises in China", *Journal of Risk and Financial Management*, Vol. 15 No. 1, p. 37, doi: 10.3390/jrfm15010037.
- Wagner, M. (2008), "Empirical influence of environmental management on innovation: Evidence from Europe", *Ecological Economics*, Vol. 66 Nos 2/3, pp. 392-402.
- Wang, Y., Abbasi, K., Yekini, K.C. and Babajide, B. (2019), "Corporate governance mechanisms and firm performance evidence from the emerging market following the revised CG code", *Corporate Governance: The International Journal of Business in Society*, Vol. 20 No. 1, doi: 10.1108/CG-07-2018-0244.
- Wang, Y., Chang, X., Wang, T. and Wang, S. (2022), "Female directors and environmental innovation: is stakeholder orientation a missing link?", *Gender in Management: An International Journal*, Vol. 37 No. 5, pp. 587-602.

- Wu, W., Alkaraan, F. and Le, C. (2023), "The moderating effects of corporate governance and investment efficiency on the nexus between financial flexibility and firm performance", *Journal of Financial Reporting and Accounting*, doi: 10.1108/JFRA-05-2023-0234.
- Xue, Y., Jiang, C., Guo, Y., Liu, J., Wu, H. and Hao, Y. (2022), "Corporate social responsibility and high-quality development: do green innovation, environmental investment and corporate governance matter?", *Emerging Markets Finance and Trade*, Vol. 58 No. 11, pp. 3191-3214.
- Yin, J. and Wang, S. (2018), "The effects of corporate environmental disclosure on environmental innovation from stakeholder perspectives", Applied Economics, Vol. 50 No. 8, pp. 905-919.
- Zaman, R., Atawnah, N., Haseeb, M., Nadeem, M. and Irfan, S. (2021), "Does corporate eco-innovation affect stock price crash risk?", *The British Accounting Review*, Vol. 53 No. 5, p. 101031.
- Zhang, L. (2012), "Board demographic diversity, independence, and corporate social performance", *Corporate Governance: The International Journal of Business in Society*, Vol. 12 No. 5, pp. 1472-0701, doi: 10.1108/14720701211275604.
- Zhu, C., Husnain, M., Ullah, S., Khan, M.T. and Ali, W. (2022), "Gender diversity and firms' sustainable performance: moderating role of CEO duality in emerging equity market", Sustainability, Vol. 14 No. 12, p. 7177, available at: www.mdpi.com/2071-1050/14/12/7177
- Zona, F., Zattoni, A. and Minichilli, A. (2013), "A contingency model of boards of directors and firm innovation: the moderating role of firm size", *British Journal of Management*, Vol. 24 No. 3, pp. 299-315, doi: 10.1111/j.1467-8551.2011.00805.x.

## **Further reading**

Dwekat, A., Seguí-Mas, E., Tormo-Carbó, G. and Carmona, P. (2020a), "Corporate governance configurations and corporate social responsibility disclosure: qualitative comparative analysis of audit committee and board characteristics", *Corporate Social Responsibility and Environmental Management*, Vol. 27 No. 6, pp. 2879-2892, doi: 10.1002/csr.2009.

#### **Corresponding author**

Aladdin Dwekat can be contacted at: aladdin.dwekat@najah.edu